Author’s response to reviews

Title: Measuring skin necrosis in a randomised controlled feasibility trial of heat preconditioning on wound healing after reconstructive breast surgery; study protocol and statistical analysis plan for the PREHEAT trial

Authors:

Suzie Cro (s.cro@imperial.ac.uk)
Saahil Mehta (saahil47@googlemail.com)
Jian Farhadi (jian.farhadi@gstt.nhs.uk)
Billie Coomber (billie.coomber@gstt.nhs.uk)
Victoria Cornelius (v.cornelius@imperial.ac.uk)

Version: 2 Date: 03 Nov 2017

Author’s response to reviews:

Dear Editor,

Please find enclosed our revised article with tracked changes. Following instructions we have reformatted the article as a protocol and SAP paper. Consequently we have added in the SPIRIT checklist and figure, restructured the abstract and edited the required section headings. Our revision now adheres to the submission guidance set out for study protocols.

Many Thanks,

Suzie Cro on behalf of the authors